Contents lists available at ScienceDirect



International Journal of Intelligent Networks

journal homepage: www.keaipublishing.com/en/journals/ international-journal-of-intelligent-networks



# A systematic review on early prediction of Mild cognitive impairment to alzheimers using machine learning algorithms



# K.P. Muhammed Niyas<sup>a</sup>, P. Thiyagarajan<sup>b,\*</sup>

<sup>a</sup> SVKM's Narsee Monjee Institute of Management Studies, Hyderabad, India
 <sup>b</sup> Rajiv Gandhi National Institute of Youth Development, Sriperumbudur, India

#### ARTICLE INFO

Keywords: Review Mild cognitive impairment Alzheimer's disease Deep learning Machine learning

#### ABSTRACT

Background: A person consults a doctor when he or she is suspicious of their cognitive abilities. Finding patients who can be converted into Alzheimer's in the future is a difficult task for doctors. A person's dementia can be converted into several types of dementia conditions. Among all dementia, Alzheimer's is considered to be the most dangerous as its rapid progression can even lead to the death of an individual. Consequently, early detection of Alzheimer's would help in better planning for the treatment of the disease. Thereby, it is possible to reduce the progression of the disease. The application of Machine Learning algorithms is useful in accurately identifying Alzheimer's patients. Advanced Machine Learning algorithms are capable of increasing the performance classification of future AD patients. Hence, this study is made on a number of previous works from 2016 onwards on Alzheimer's detection. The aspects such as the country of the participants, modalities of data used and the features involved, feature extraction methods used, how many follow-up data were used, the period of Mild Cognitive Impairment to Alzheimer's Disease converters predicted, and the various machine learning models used in the previous studies of Alzheimer's detection are reviewed in this study. This review helps a new researcher to know the features and Machine Learning models used in the previous studies for the early detection of Alzheimer's. Thus, this study also helps a researcher to critically evaluate the literature on Alzheimer's disease detection very easily as the paper is organized according to the various steps of the Machine Learning process for Alzheimer's detection in a simplified manner.

# 1. Introduction

Alzheimer's is one of the deadly dementia as the rate of progression of the disease is rapid. It can even lead to the death of an individual. Alzheimer's can destroy the cognitive abilities of an individual in a dangerous manner [1]. Usually, a person consults when he or she suspects his or her cognitive abilities. The cognitive impairment of a patient can lead to several types of dementia. Among various types of dementia, it is difficult for a doctor to find out the patients who will have Alzheimer's in future [2].

Finding the crucial bio-markers for determining Alzheimer's and other type of dementia are a crucial investigating area among the researchers [3–5]. The study conducted by the researchers in Refs. [3–5] experimented on a rat model for the detection of Alzheimer's patients. The main idea behind the experimentation on rat models is to observe how effective the drug is in reducing the progression of Alzheimer's disease. Hence, much of the latest research mainly revolves around the

experimentation of Alzheimer's symptoms and medications on rats [3–8]. All of these studies focus on observing the risk factors of AD among the rats which the doctors believe could be replicated among human beings also. Hence, the physicians are using the neuroimage changes that were observed among the rats for conducting a study on the human body also [3–8].

Finding Alzheimer's patients at the baseline or screening visit itself helps doctors in making better health related decisions. Thus, the doctor can plan better medication strategies that reduce the progression of the disease. Further, early detection of Alzheimer's helps in better health related outcomes of the patient [9]. The treatment strategies are also more effective during the early stages of the progression of the disease. Likewise, the effect of drugs is also more effective during the early stages of the disease [9].

It is also a challenging task to find out patients who will convert to Alzheimer's in future. This is because the exact parameters responsible for conversion of Mild Cognitive Impairment (MCI) to Alzheimer's

\* Corresponding author. *E-mail addresses:* muhammedniyas350@gmail.com (K.P. Muhammed Niyas), thiyagu.phd@gmail.com (P. Thiyagarajan).

https://doi.org/10.1016/j.ijin.2023.03.004

Received 20 July 2022; Received in revised form 28 March 2023; Accepted 29 March 2023

Available online 7 April 2023

<sup>2666-6030/© 2023</sup> The Authors. Published by Elsevier B.V. on behalf of KeAi Communications Co., Ltd. This is an open access article under the CC BY-NC-ND license (http://creativecommons.org/licenses/by-nc-nd/4.0/).

Disease (AD) still has not been found out. Generally, the doctors take this decision by analysing data such as Magnetic Resonance Imaging (MRI), Positron Emission Tomography (PET) and Neuro-psychologicalassessments data. But, this is challenging to the doctors as the characteristics of MCI to AD converters is very complex and hard to trace. In such a scenario, a predictive Machine Learning (ML) model assists in good decision making process [2,10]. Researchers have focused mostly on longitudinal (visit) data of patients for early prediction. Longitudinal data collected over various follow up periods say first, second, third visits and so on contain the relevant information and parameters that would progress over time [11].

Some patients who have MCI would convert to AD in future namely Progressive Mild Cognitive Impairment (pMCI). However, some patients may remain with MCI without converting to severe AD in the future namely Stable Mild Cognitive Impairment (sMCI). As far as a physician is considered, it is important to find out the future sMCI and pMCI patients [2,10,11]. If the physician can detect future AD patients in advance, then it is possible to recommend the proper diagnosis and medication strategies for an AD patient. This would help in reducing the progression of the dangerous AD to the next level where the patient and their relatives would find it very difficult to deal with. In the same manner, the detection of sMCI also helps a physician to prescribe the required medication for a patient based on his or her cognitive impairment. This would help in saving the unnecessary costs that might spend on the MCI patients [2,10,11].

The biggest challenge in distinguishing between pMCI and sMCI patients are as follows:

- Overlapping of certain characteristics is seen in both pMCI and sMCI patients. For example, the shrinkage of brain regions is commonly seen among both pMCI and sMCI patients [2].
- It is very difficult for a domain expert to find out the crucial biomarkers and features that are responsible for distinguishing pMCI and sMCI. Thus, a physician seeks help from an advanced Machine Learning (ML) technique for deriving out the hidden patterns and relevant features that are responsible for distinguishing the pMCI and sMCI [2,10,11].
- Identifying the crucial bio-markers for distinguishing pMCI and sMCI patients from unstructured data like medical images is a major challenge [2].
- The inability to generalize the characteristics from various multimodalities of data is also a major challenge [2,10,11].

As it is challenging to identify and distinguish the sMCI and pMCI patients by observation even for an experienced practitioner, a ML algorithm would help in assisting the physician in properly identifying the AD risk patients. The ML algorithms are capable of identifying the distinguishing features that can differentiate a sMCI from a pMCI patient. It is difficult for the physician to generalize or classify a patient as sMCI or pMCI as many of the characteristics and symptoms are common for both sMCI and pMCI patients [2,10-12]. This is where an efficient ML algorithm would be very helpful as it can identify the hidden patterns in the complex AD data of the patients using a mathematical model. Consequently, an efficient ML model can act as a good assist for the physician while making crucial decisions to predict whether a patient will have a sMCI or a pMCI [11-13]. A workflow of ML algorithm for sMCI and pMCI patient includes: 1. Identifying the important features from a huge repository of structured or unstructured dataset, 2. Classifying the patient as sMCI or pMCI using longitudinal visit data.

The above challenges resulted in ML researchers working on various neuro-imaging data, cognitive tests, and other demographic data for model building. It is a major challenge among ML researchers who are working in AD detection to generalize a specific set of bio-markers from unstructured and structured datasets for distinguishing pMCI and sMCI patients. Hence, the ML researchers have already explored various methods for distinguishing pMCI and sMCI patients. This research paper helps a new researcher who is entering MCI to AD detection using multimodal data in giving a simplified overview of the previous studies. Many researchers have already adopted various types of features and classifiers for detecting MCI to AD converter patients. This paper is a summarized review of the various methods and techniques that are used for identifying MCI to AD converters. Consequently, this study will help an early researcher to find out their research direction in the classification of pMCI and sMCI patients.

Hence, key focus on early prediction of MCI to AD converters using ML models is discussed in this paper. A systematic review of identified 42 research papers are taken into consideration and it is summarized to answer the research questions. The key focus of review is towards the papers implemented longitudinal data in prediction of MCI to AD. The summarized data consists of the study participant's country, features, follow up data characteristics, classifiers and performance metric that are implemented in the Alzheimer's detection. These aspects are explained in detail for every selected study.

This study used a review methodology for finding out how the previous studies addressed the MCI to AD conversion using ML techniques. The literature review is conducted on previous studies from 2017 onwards for MCI to AD conversion. The proposed approach of the review paper involves a detailed study of 42 state of the art articles on MCI to AD conversion.

This paper is organized as follows: section 2 contains the review protocol and methodology, section 3 contains results and discussion, section 4 contains the limitation of the study, section 5 contains the conclusion.

#### 2. Review protocol and methodology

This section contains a detailed explanation of the method of review protocol used in the paper. Initially, a set of research questions are formulated. The answers to the research questions are found after examining the previous literature that deals with the detection of AD using ML techniques. Initially a search strategy is followed based on the research questions. Then, the data from the selected studies are extracted for every research question.

#### 2.1. Research questions

The overall aim of this research paper is to provide an overview of the current state of the art in early detection of MCI to AD converters using machine learning approaches. For providing an overview of the various techniques in AD detection, the following research objectives are framed. The following research objectives are used to address the aspects such as the country of the study participants, the various data modalities used, ways of extracting features from unstructured neuroimage data and classifiers used in predicting sMCI and pMCI patients.

- RQ1:What is the country of the study participants?
- RQ2:What modalities of data are used as features?
- RQ3:Which papers used feature selection techniques for ranking the features?
- RQ4:How many follow up data from longitudinal data are used? What is the duration of follow up data?
- RQ5:What range of MCI Conversion to AD conversions are found out in the papers?
- RQ6:How many papers used machine learning and deep learning models?
- RQ7:Do ensemble models improve the performance in prediction?
- RQ8:When do the models perform better?

RQ1 is the basic question for finding out whether a model is developed for predicting MCI to AD converters from various parts of the world, as AD and MCI can vary in their intensity and causes from region to region depending upon multiple factors [14,15]. The relevance of this question is to identify whether prediction models are developed in a generalized population, say persons belonging to various regions or countries.

RQ2 concerns the features that are used for prediction. The possible type of features can be from single modality or multiple modalities of data. The most common modalities of data are MRI, PET, functional MRI (fMRI), Electroencephalogram (EEG), Neuro-psychology data, Clinical data and Cerebro-Spinal Fluid data. Here, we search the answer for what type of features are used from various types of modalities. The question is relevant because the parameters that are responsible for MCI to AD conversion are not specifically discovered for distinguishing MCI to AD converters [16].

RQ3 aims to find out if researchers used any feature selection techniques for getting relevant and important features. This is because the exact features responsible for conversion of MCI to AD is not known among the researchers. Therefore, it is important to find out and rank the important features for classifying MCI to AD converters [9,10].

RQ4 aims to find out the number of follow up data used by researchers on the longitudinal classifiers for distinguishing MCI to AD converters. Further, the duration between the follow-up data is also observed for the papers. This is an important research question because it is important to investigate whether the researchers are able to classify the MCI to AD converters within shorter follow up period.

RQ5 investigates the effectiveness of prediction models in finding out the MCI conversion to AD. If we can find out MCI to AD converters early, it will help in a good decision making [9]. Therefore, this question will explore the range of conversion of MCI to AD converters from the literature.

RQ6 explores the various longitudinal models used for the classification of pMCI and sMCI patients. The objective of this question is to explore the different types of longitudinal models used for finding sMCI and pMCI patients and to explore which models achieve better performances. The various performance metrics such as Accuracy, Balanced Classification Accuracy (BCA), Sensitivity and Specificity are assessed to understand the classification performance for distinguishing the pMCI and sMCI patients.

RQ7 aims to explore the ability of ensemble models in improving the prediction results. This question is investigating whether ensemble models are good in improving the results in terms of BCA, Sensitivity, Specificity for distinguishing pMCI and sMCI patients.

RQ8 aims to explore what kind of features and classifier models are improving the classification performance for pMCI and sMCI patients.

#### 2.2. Search strategy

PRISMA guidelines are followed for selecting the review papers [17]. The databases like Google Scholar, IEEE Xplore, Scopus, and Pubmed are searched for selecting the relevant studies. The searching for relevant studies is conducted by using specific keywords and phrases. We have used phrases such as "MCI to AD Converters", "Early Detection of MCI to AD Conversion using Machine Learning", "Early Detection of Alzheimers", "Alzheimers Converters" in the journal search engines. The records identified by searching as on 31/08/2022 with the keywords in each journal are given in table 1.

#### 2.3. Study selection

To select the relevant studies, an inclusion and exclusion criteria is developed. They are used in the various stages of the study selection process. The following inclusion and exclusion criteria is followed in this paper.

#### 2.3.1. Inclusion and exclusion criteria

After getting the results from the databases for the keywords as mentioned in Table 1, the duplicate papers are removed from the results. Initially, the papers are selected after screening the abstracts. The papers

## Table 1

Number of records found for each query in journal search engines as on 03/10/2022.

Search Engine	Keywords	Number of Records Found		
Google	MCI to AD Converters	9970		
Scholar	Early Detection of MCI to AD Conversion Using Machine Learning	5250		
	Early Detection of Alzheimers	4660		
	Alzheimers Converters	7545		
IEEE Xplore	MCI to AD Converters	2345		
-	Early Detection of MCI to AD Conversion Using Machine Learning	1610		
	Early Detection of Alzheimers	1750		
	Alzheimers Converters	1992		
Scopus	MCI to AD Converters	1200		
	Early Detection of MCI to AD Conversion Using Machine Learning	1809		
	Early Detection of Alzheimers	1610		
	Alzheimers Converters	1890		
Pubmed	MCI to AD Converters	5500		
	Early Detection of MCI to AD Conversion Using Machine Learning	5610		
	Early Detection of Alzheimers	5890		
	Alzheimers Converters	5100		

with no proper explanation about the ML techniques in the abstract are excluded from the study. Then, the inclusion and exclusion criteria is implemented on the resultant selected papers.

The papers are selected on the basis of inclusion and exclusion criteria.

- The papers that used only longitudinal data for the research are selected for the study. The reason is that the longitudinal studies analyze the follow-up data of the patients for making the final decisions about the classification of pMCI and sMCI patients. Hence, the studies with only cross-sectional data for the classification of pMCI and sMCI patients are eliminated from the study.
- The papers that used only longitudinal data for the research are selected for the study because it analyses the follow-up data of the patients for making the final decisions about the classification of pMCI and sMCI patients. Hence, the studies with only cross-sectional data for the classification of pMCI and sMCI patients are eliminated from the study.
- The papers with clear explanation about the performance metrics are used for the study. The studies with clear explanation about any of the performance metrics such as Accuracy, BCA, Sensitivity and Specificity are selected for the systematic review.

### 2.4. Data extraction

The relevant information for every paper needs to be extracted in this stage. The following information is extracted from each paper that will answer the following research questions: country of study participants, the number of follow-up longitudinal data used, range of MCI conversion predicted, description of the modalities of data, is automatic feature extraction used or not, type of algorithm used (machine learning, deep learning), ensemble learning and the Sensitivity, Specificity, and Accuracy reported in each of the finally selected papers. As far as the results are concerned, the best results in each paper (in terms of Accuracy, Sensitivity, and Specificity) are chosen as the answer to the research questions. The research papers contain many experiments (by varying features, follow-up data, and model combinations), and the best out of the experiments (in terms of Accuracy, Sensitivity, and Specificity) is chosen as the result of our study.

#### 3. Results and discussion

This section contains detailed information about the results and discussions.

#### 3.1. Study selection

In this section the overall study selection procedure is explained in detail. Searching in the journal repositories like IEEExplore, Google Scholar, Pubmed, and Scopus are reported with 15450 results. 735 duplicate papers are removed from the reported results. Then, the abstracts are screened for the resulting 14715 papers. There was an exclusion of 3343 articles after abstract screening. Then, the papers without any Artificial Intelligence applications which explained only the scientific aspects of MCI and AD are excluded at the abstract level screening. Among the 11372 articles assessed for eligibility, there was an exclusion of the papers that do not have MCI to AD converters (n = 3356), longitudinal data (n = 4200), specified either method or features (n = 3513), performance metrics description including Accuracy, Sensitivity, Specificity (n = 391). Consequently, 42 studies were selected for conducting the systematic review. Fig. 1 contains the exclusion criteria used in the paper selection process.

Following are the public datasets used for predicting pMCI and sMCI patients from the literature.

Public datasets for predicting the pMCI and sMCI patients are as follows:

- Alzheimer's Disease Neuro-Imaging Initiative (ADNI) Dataset: ADNI dataset is used for uniting various researchers who are working on the progression of MCI to AD. ADNI project consists of multimodal data belongs to MRI, PET, neuro-imaging, genetic and other cognitive test data [18–20]. The dataset consists of participants' data from North American hospitals collected over 5 years. The main aim of the research project is to examine and find out the bio-markers that are responsible for the early detection of Alzheimer's Disease using Machine Learning algorithms [18–20]. ADNI datasets are developed for optimizing, validating, standardizing clinical trial measures and biomarkers developed for clinical research [18,19].
- Australian Imaging Biomarkers and Lifestyle Study of Ageing (AIBL) dataset: AIBL dataset consists of 18-month follow-up data of Australian patients ranging from 50 to 90 years. The dataset consists of MRI, PET, cognitive test and lab test data of the patient which is collected over 1 year. The primary objective of the AIBL dataset is to find out which cognitive tests, medical imaging characteristics, and lab tests are crucial in distinguishing MCI and AD patients [20–22]. It consists of a multidisciplinary research team from researchers located in Perth, Sydney, Melbourne, Canberra, and Brisbane [20–22].
- Open Access Series of Imaging Studies (OASIS) dataset: OASIS dataset consists of multimodal of more than 1000 patients collected over various points of time. The dataset consists of neuro-imaging, cognitive tests data that are collected over a period of time. The dataset consists of 3D sliced MRI images of more than 10000 images. The objective of the project is to find out the future MCI and AD patients in advance using clinical, neuro-imaging, genetic and other biomarkers [23,24]. The OASIS dataset consists of patient's data collected from 416 subjects aged from 16 to 96 years. The subjects of the study are right handed. Most of the samples of the dataset are more than 50 years old [23,24].
- Dementia Bank clinical dataset

Dementia Bank clinical dataset consists of Alzheimer's and Healthy Control patient data from the speech data. It consists of dementiarelated data of 500 patients collected over various time points [25, 26]. The data set consists of transcripts and video recordings of AD and healthy patients. This data set consists of speech utterances of the AD



Fig. 1. Exclusion criteria used in the paper selection process.

and HC patients that were observed after performing a recall test on multiple scenes [25,26].

#### 3.1.1. RQ1:What is the country of study participants?

- Of the included studies, ADNI dataset is used in many studies [18, 19]. The participants of the dataset are from various parts of USA and Canada. The papers that experimented on ADNI dataset are: [27–62]. All the studies mentioned above were conducted on study participants from North America. 3 studies were conducted on participants from Europe [50,54,55,63,64].
- One study contains study participants each from Portugal [63] and Germany [64].
- Inter continental study on participants from Milan in Italy is mixed with the ADNI dataset in Ref. [65].

3.1.2. RQ2: What are the modalities of data used as features?

Our main focus on this research question is to find out the papers which used single modality and multiple modalities of features for the prediction purposes.

#### • Single Modality:

Data from a single modality alone is used for the prediction purpose in many studies.

MRI: Hippocampal volume is selected as feature in Refs. [27,35]. Free Surfer Intensity values of selected regions are used by the researchers [44,49]. Entorhinal Volume is used by the researchers in Ref. [27]. Amygdala distance is used in the experiments [36]. Further, Intracranial volume is also used by the studies [27]. Cortical thickness is also used in the study [34,48]. Distinguishing Voxels of brain MRI are selected by the researchers using Logistic Regression for longitudinal training data in Ref. [38]. Gyrus height is again another feature used by the researchers in Ref. [48]. A network based feature is created for a purely MRI based feature selection technique by Ref. [48]. Whole patches of MRI image is used as a feature in 1 study [29]. The whole 2D slice's intensity features of a brain MRI image is used by the researchers in Ref. [38] for distinguishing sMCI and pMCI. A transfer learning approach is employed in the whole slices of MRI image for distinguishing the pMCI and sMCI patients [54]. The transfer learning approach used by the researchers in Ref. [54] is focused on hyperparameter tuning of convolutional neural network on the basis of age. An attention based neural network architecture is developed by the researchers in Ref. [66] for predicting MCI to AD conversion patients.

**PET**: PET data alone is used in Ref. [31]. Metabolic intensity values are extracted from the raw PET images for all patches by the researchers in Ref. [31]. PET imaging techniques are widely used by the researchers for early detection of Alzheimer's disease using convolutional and recurrent neural networks [67]. There were lot of studies conducted for finding out the difference in PET feature like florbet-pair for distinguishing AD and healthy control patients. PET imaging techniques are widely used by the researchers for finding out the Alzheimer's disease patients from the past. PET imaging technique is capable of finding out the metabolic features in the brain. For instance, the cognitive memory is high for people with good metabolism inside the brain [31, 67].

**Electroencephalogram (EEG)**: EEG data is used by the researchers in Ref. [52]. They created a graph structure for the functional connectivity in the EEG signals of sMCI and pMCI patients [52]. Researchers also explored the Time frequency, Bi-Spectral features of EEG data for accurate prediction of MCI to AD converters [55].

#### • Multiple Modality:

From the selected papers, in 12 papers researchers experimented with multiple modalities of data say MRI, PET, Neuro-psychological and clinical data. A Combination of unstructured data from various modalities say that MRI, PET and clinical data are used by the researchers in Ref. [34]. They used Temporal Gyrus and Hippocampus volumes from MRI, Independent Component Analysis (ICA) from PET and MMSE values as neuro psychological assessment data [34]. A combination of MRI and PET is used by the researchers in Ref. [37].

A combination of functional MRI (fMRI) and MRI is used in the study [46]. A connectivity matrix for 93 ROI is created as a feature and the features of FREESURFER software are extracted from MRI [46].

A combination of structured data from various modalities is used by the researchers in 4 studies [40,42,50,63]. Both MRI and Neuro-psychological data are used by the researchers in Ref. [40]. They used Hippocampus, Ventricles, Fusiform as MRI data, Alzheimer's Disease Assessment Scale-cognitive subscale (ADAS-Cog) and MMSE as neuro-psychological data in the study. Another study [63] used both neuro-psychological and clinical data for the prediction purpose. Neuropsychological scores such as MMSE, ADAS-Cog and clinical data such as demographics and cardiovascular scores are used in these studies. Both MRI and CSF data are used in Ref. [41]. Hippocampal Volume is used as MRI feature and A-Beta data is used as CSF features in this study [41]. Researchers used both structured MRI and CSF data for prediction purpose [64]. MRI data consists of Hippocampal volumes and Tau, A-Beta are used as CSF features in the study [64]. Researchers in Ref. [50] used socio-demographics data such as age, sex, years of education, marital status and psychological test data such as MMSE, ADAS-Cog, Rey Auditory Verbal Learning (RAVLT) in the experiments. A health ageing trajectory model was developed on multimodal structured dataset by the researchers in Ref. [58] for predicting MCI to AD converters.

A hybrid of structured and unstructured data from various modalities are used in 5 studies [34,41,43,45,47,53]. Unstructured MRI, PET and neurological, clinical data are used by the researchers in Ref. [34]. The study includes MRI data of voxel based morphometry and volume based morphometry, neuro-psychological data as MMSE and clinical data as demographics medical history data [34]. In another study, a combination of MRI, PET, CSF, Demographics and Neuro-psychological data is used by the researchers in Ref. [43]. Similarly, a mix of MRI, Genetic and Neuro-psychological data is used by the researchers in 2 studies [45,47]. Atrophy score of each MRI, Polyhazard Score is used as a genetic score and MMSE Neuro-psychological data is used as features in the study [45]. In a similar study, researchers used texture value of voxels of MRI, APOE4 Genetic data and MMSE as neuro-psychological data [47]. The texture value of voxels are captured using neural networks. Texture values capture the Gray and White matter intensity inside the brain. These features are combined with the APOE4 genetic data and the most commonly used cognitive score namely MMSE [47]. In another study, Unstructured MRI, PET and CSF are used as features for sMCI and pMCI. MRI data consists of Gray Matter pixel values of 93 ROIs. These 93 ROIs are captured in the three views of neuroimage namely coronal, sagittal, transverse. It is combined with CSF data which consists of TAU and A-Beta protein levels [41]. The Ventricular Cavity and reduction of Gray Matter volumes are extracted from the MRI images by the researchers in Ref. [53]. They also added the cognitive test Mini-Mental State Examination (MMSE) for predicting the sMCI and pMCI patients [53]. The whole patches of brain image and cognitive assessment data is used by the researchers in Ref. [56] for sMCI and pMCI converters.

#### 3.1.3. RQ3: Which papers used automatic feature extraction methods?

Some papers used automatic feature extraction methods for extracting relevant information. The feature selection methods are employed in unstructured data like MRI, PET, EEG.etc. Whereas some papers do not have any feature extraction methods. The papers taken for survey are classified based on feature extraction as given below.

#### - No Feature Extraction

There are 15 studies without any feature extraction methods. These papers used all the structured features that are already derived from the existing literatures [27,33–36,40–42,47,49,50,63,64].

#### - Automatic Feature Extraction

16 studies applied automatic feature extraction techniques on unstructured data such as MRI, PET, and EEG [28–33,37–40,43–46,48,51, 65]. Following are the various techniques used for feature extraction from MRI, PET, and EEG data.

#### 3.2. MRI

It is observed that the researchers were mainly focused on using deep neural network architectures for extracting features from MRI data either on entire images or focused areas of the brain. The studies that considered the whole patches of the brain for automatic feature extraction are as follows:

Whole Patches of the brain: A deep learning method combined with a sparse regression model is used for predicting sMCI and pMCI patients [28]. A deep CNN is used for extracting the neuro-image features which is then combined with a sparse regression model for accurate classification of sMCI and pMCI patients [28]. A multi-domain transfer learning based approach is used by the researchers for classifiying the sMCI and pMCI patients [32]. Transfer learning is applied on various kinds of slice image for classification of sMCI and pMCI patients. A different approach was used by the researchers in Ref. [33] where a graph based network is extracted from the neuroimages and graph based measures are extracted for distinguishing sMCI and pMCI patients. A CNN based architecture is developed for the early conversion of sMCI patients by the researchers in Ref. [38] for predicting 1 year MCI to AD conversion patients. A supervised non-parametric method is implemented by the researchers in Ref. [40] on longitudinal data for sMCI and pMCI prediction. The study conducted by the researchers in Ref. [43] used a deep neural network method combined with an automatic feature selection technique for classifying sMCI and pMCI patients. CNN is used by the researchers in Ref. [44] for classifying sMCI and pMCI patients. An ensemble of CNN is used by the researchers in Ref. [65] for the classification of pMCI and sMCI patients. Researchers also implemented an age adjusted neural network for predicting MCI to AD patients. The neural network has an age adjusting layer which is capable of differentiating the sMCI and pMCI patients [54]. A CNN neural network architecture is implemented for extracting the GM intensity from the MRI images [56]. A variational auto-encoder neural network architecture is used for extracting the image intensity features from MRI images [57]. High dimensional MRI images are processed using CNN for the detection of sMRI and pMRI patients in the study conducted by the researchers in Ref. [56]. The researchers also developed a novel neural network architecture for the detection of sMCI to pMCI patients using neural network architectures [51].

Transfer learning approach with 3D ResNet architecture utilizing a zero shot learning is implemented by the researchers in Ref. [66]. The advantage of this method is that it is implemented on less number of MRI image data but achieved a higher classification performance for sMCI and pMCI detection. Again researchers applied a transfer learning method for extracting the intensity features from MRI images using an attention based network for capturing the functional connectivity of the brain.

The ensemble classifiers are used by the researchers in Ref. [65] for the accurate prediction of sMCI and pMCI patients. They used an ensemble of many CNN classifiers on the entire MRI brain image.

**Domain Specific Regions:** A deep learning architecture is developed on a specific Region of Interest (ROI) by the researchers in Ref. [29] where a multi-instance based deep learning framework is developed for extracting the features. The relevant features from 91 ROI's are extracted using multiple instance based deep learning architecture. Karuppi et al. [45] used a feature extraction technique focused on finding the structural volume of MRI data from MRI images. In another study, the researchers extracted morphological features from the MRI data on 91 ROI that are relevant in classifying sMCI and pMCI patients [48].

## 3.3. PET

The study conducted by the researchers in Ref. [30] used a multi-scale approach for the diagnosis of Alzheimer's disease detection. They used a CNN that can extract multiple features from multiple sub-regions of the PET image. They considered all the sub-region features of the PET image at various slices for feature extraction [30].

#### 3.4. Multimodal data

A multimodal image fusion is performed by the researchers in Ref. [33] for the early detection of sMCI and pMCI patients. A hierarchical framework was developed by the researchers in Ref. [39] for selecting the relevant features that are responsible for classifying sMCI and pMCI patients. An automatic feature extraction technique that extracts the functional connectivity using EEG data within brain regions accompanied by cognitive tests is implemented by the researchers in Ref. [52]. A novel multimodal EEG data based study was conducted by the researchers in Ref. [55] for automatic feature extraction. They extracted High Order Statistics from the Bi-Spectrum data for distinguishing sMCI and pMCI patients. The CNN is used by the researchers for extracting neuro-image intensity features from the MRI data [37]. A combination of cognitive tests and lab test data is also used by the researchers for classifying sMCI and pMCI patients [37].

An integration of MRI and Functional MRI is made by the researchers in Ref. [46] for predicting the sMCI and pMCI patients. They used neural network architectures for classifying sMCI and pMCI patients.

# 3.4.1. RQ4:How many follow-up data from longitudinal data are used? What is the duration of the follow-up data?

Follow-up time period of a patient refers to the frequent time periods in which a patient's conditions are monitored in a hospital or by a doctor.<sup>1</sup> The 3 follow-up duration used by the researchers for prediction of MCI to AD converters are 6 month, 12 month and 1 year respectively. A detailed explanation of the follow up duration and the number of such follow-up (visits) used by the researchers are explained in this section.

#### - 6 MONTHS FOLLOW-UP:

6 months follow-up are examined in the following studies: [27–48, 50,59].

Following is the information about how long the 6 month follow-up data is used by the researchers. One study contains the complete 1.5 year follow-up data of patients [28]. The complete 2 year follow-up data is used in the study [31]. The complete 2.5 years data is used in the study [42],

Among these complete 3 year follow-up, 6 month duration data are used by the researchers in Refs. [27,29,30,32–34,36,37,39,41,43–46, 48–50,65]. The complete 3.5 years follow-up data is used in Ref. [35]. However, the complete follow-up data till all 3.5 years is not used in this study. The complete 4 years data is used in Refs. [38,47].

- 3 years, 12 month:

The study conducted by the researchers in Ref. [64] contains follow-up data of 12 months duration and it is followed for 3 years (number of follow-up is 3).

- 1 year, 5 month:

The study conducted by the researchers [63] contain follow-up data of 1 year duration and it is followed for the next 5 years (number of follow up data used is 5).

– 3 years

Follow-up data is 2 years for [61].

- Unclear follow up data

<sup>&</sup>lt;sup>1</sup> https://www.cancer.gov/publications/dictionaries/cancer-terms/def/follow-up.

The follow-up data is unclear for the studies in: [51–58,60,62].

#### 3.4.2. RQ5:What range of MCI conversion to AD is used?

Range of MCI to AD conversion is very vital in proper health care decision making processes. The description of time duration of MCI to AD conversion ranges are explained below.

- Six Month:

The shortest conversion time span in the literature is 6 months [39, 49].

- One Year: 4 studies predicted 1 year conversion from MCI to AD [30, 37,38,64].
- One and Half Year: 2 studies experimented on 1.5 year conversion prediction of MCI to AD [28,40].
- Three Year: Most of the studies experimented on prediction of 3 year conversion of MCI to AD [29,31–34,36,41–43,45–48,50,65].
- Five Year: Longest time span of prediction of MCI to AD is 5 years which is experimented in the paper [63].
- No Conversion Range (Just MCI to AD Converters Without Time Period Specification). There was no specific time of conversion from MCI to AD in the studies [35,44]. Both studies predicted whether a person will have conversion from MCI to AD without finding the time of conversion.
- The MCI to AD conversion range is unclear in the following studies: [49–62,65].

# 3.4.3. RQ6:How many papers used machine learning and deep learning separately?

A classification of papers which used machine learning and deep learning techniques is as given below.

- ML algorithms:

Following ML algorithms are used:

SVM is widely used by researchers as a final classifier for predicting the sMCI and pMCI detection. Initially, the deep neural networks such as CNN or different types of CNN are used for extracting the relevant features from the unstructured data. Then, the extracted features are fed into the SVM classifier. SVM is used in the given studies: [34,36–40,42, 43,47,48,64].

LASSO is used for both the feature selection and classification purpose in the study conducted by the researchers in Ref. [31]. A feature selection framework is built for selecting the relevant features and a classifier is implemented to classify the sMCI and pMCI patients [31]. In addition, a linear Sparse Regression model is developed by the researchers in Ref. [41].

The Cox based survival event time detection models are used by the researchers in Refs. [33,45]. A Cox regression model is used as a classification model for predicting the sMCI and pMCI patients using the unstructured data collected by CNN models in the study [33]. Similarly, a Cox hazard model is implemented by the researchers in Ref. [45] for classifiying sMCI and pMCI patients. This model helps in identifying whether a patient belongs to sMCI or pMCI category on the basic Cox probability value [45].

Researchers also used longitudinal classifiers for the detection of sMCI and pMCI patients [35,63] for detecting the sMCI and pMCI patients. A Mixed Effects Model is proposed by the researchers in Ref. [63] for the predicting sMCI and pMCI patients. This model considers all the varying visit intervals of the patients. Moreover, a sliding window based approach is also used for the classification of sMCI and pMCI patients [35]. This model is capable of measuring the influence of one visit over the other in predicting a patient as sMCI or pMCI.

An ensemble of Machine Learning classifiers is also used by the researchers [27,50]. An ensemble of SVM and Logistic Regression is used by the researchers in Ref. [27] for sMCI and pMCI detection. Likewise, an ensemble of SVM and LR is used by the researchers in Ref. [50] for distinguishing sMCI and pMCI patients. Table 2 contains the purpose and the available configuration information about the ML algorithms used in the study.

- Deep Learning:

Deep Learning neural network architectures are used by the researchers as classifiers as well as feature extractors by the researchers in Refs. [28,30,44,46,49,52–62,65]. A CNN architecture is used for

#### Table 2

Configuration of ML algorithms used in previous works. RBF-Radial Basis Function, FCL-Fully Connected Layers, BLR: Binary Logistic Regression.

Reference	ML Configuration	Purpose
[27]	SVM: RBF kernel	Classification of AD based on
		structural features
[28–30,47] [50,	CNN: 8 FCL, Sigmoid	Capturing Gray and White Matter
54,66] [31]	(Activation) SAE: Rectified Linear	Intensity Capturing Gray and White Matter
[31]	Unit	intensity
[32]	CNN + SVM	CNN: Metabolic Intensity Extraction
[01]	CNN: 3D with Sigmoid	SVM: Classification of sMCI, pMCI
[33,35,36,39,	CNN + SVM	CNN: Gray and White Matter
65]		Extraction
	CNN: Rectified Linear	SVM: Classification of sMCI, pMCI
504.447	Unit	
[34,44]	Cox Regression Models	Calculating AD survival from
[37]	SVM	multimodal data Classification of sMCI, pMCI
[37]	5 V IVI	patients
		Using Amygdala distance
[38]	SAE + SVM	SAE: Capturing intensity from
		multimodal neuroimages
	SVM: RBF Kernel	SVM: Classification of pMCI, sMCI
		patients
[41]	LR	Selection of significant Voxels
[40,61]	CNN, SVM, SVR	classifying sMCI, pMCI CNN: Automatic Feature Extraction
[40,01]	GININ, SVIN, SVIC	SVR: Estimating mean Gray and
		White matter
		SVM: Classification of sMCI, pMCI
[42]	Sparse Learning	Causal Inference Model
	Regression	Identifying relation between
		features
[38,43,46,59,	SVM: Polynomial Kernel	Classifying pMCI, sMCI patients
64] [45,52]	SVM: RBF Kernel	Classifying pMCI, sMCI using graph
[40,02]	SVW. RDF Reffici	measures
[63]	CNN + BLR	CNN: Undefined FCL
		BLR: classification of pMCI, sMCI
		patients
[48]	Bayes Classifier	Probabilistic model for speech
5 4 6 7		features
[49]	CNN + Graph networks	CNN: Automatic Extraction of
	CNN: Activation	neurimage features Graph Network: Extracting
	function as tanh	Connectivity measures
[60]	SVM, KNN, MLP	Ensemble Voting Classifier
	SVM: RBF Kernel	-
	KNN: 5	
	MLP: Undefined	
[53]	CNN with attention	Focusing on ROI more
[55]	mechanism CNN	Extracting features from EEC data
[55] [66]	CNN + SVM	Extracting features from EEG data CNN: Extracting neuroimage
[00]		intensity
	SVM: RBF kernel	CNN + SVM: Classifying sMCI,
		pMCI patients
[58]	Multi-Kernel SVM	Classifying pMCI, sMCI patients
	2 RBF kernel	
[44.40.65]	1 Polynomial kernel	
[44,49,65]	CNN + RNN	CNN: Automatic Feature Extraction
-		RNN: Capturing Temporal features

extracting the neuroimage intensity features such as Gray Matter and White Matter intensity by the researchers in Refs. [28,30,46,52–55, 58–62,65].

Similarly, a combination of CNN and Recurrent Neural Network (RNN) is used by the researchers in Refs. [44,49,65]. These models are capable of extracting the spatial intensity based features using the deep CNN architectures. Then, the deep RNN architectures are used for extracting the temporal information from the extracted features of the unstructured data [44,49,65].

# 3.4.4. RQ7:Do the ensemble models improve the sMCI and pMCI prediction?

Ensemble learning algorithms are used in Refs. [27,29,33,39,40,44, 49,50,50,65]. The ensemble deep learning architectures such as CNN-RNN combination are used by the researchers in Refs. [44,49,65] for capturing the spatio-temporal features from the neuro-images and expected to increase the classification performance. Moreover, the ensemble of classifiers is used for classifying whether a patient is s MCI or pMCI is also expected to increase the classification performance of detecting sMCI and pMCI with a decision making approach that captures the diverse information [29,33,39,40,50].

The range of accuracy reported using ensemble models is 74%–89%. The range of sensitivity reported using ensemble models is 54%–89%. The range of specificity reported using ensemble models is 73%–92%.

On comparing the overall performance of the models, it is observed that the ensemble models have not achieved the best results. In fact, ensemble models give better results but not the best ones. However, this again emphasize towards the importance of finding relevant features.

#### 3.4.5. RQ8:When do the models perform better?

In this section, the range of values reported for various performance metrics such as accuracy, sensitivity and specificity is analysed and analysis of what type of features and models are good in performing better results is given.

- Best Accuracy, Sensitivity and Specificity:

A combination of fMRI and MRI data features has achieved a higher accuracy, sensitivity and specificity of 96%, 94% and 100% respectively on ADNI dataset followed for 3 years within a time duration of 6 months.

#### - Overall Performance

Overall, the range of accuracy, sensitivity and specificity reported are 65%–96%, 54%–96%, 53%–100% respectively as given in table 3. As far as single modality is concerned, the range of accuracy, sensitivity and specificity are 71%–92%, 42%–96%, 53%–92% respectively as given in table 3. As far as multi-modality of data is concerned, the range of accuracy, sensitivity and specificity are 74%–96%, 54%–94%, 83%– 100% respectively as given in table 3. From the above results, in general, it is clear that the models performed comparatively better with multiple modalities of data.

### - Parameter Performance within modalities

The range of accuracy, sensitivity and specificity reported for MRI data alone is 65%–83%, 69%–95% and 53%–90% respectively as given in table 3. Free Surfer intensity values experimented by Refs. [44,49] has shown better performance in the result as given in table 3. The accuracy, sensitivity and specificity for the study [31] reported for PET data alone is 83%, 87% and 78% respectively. The extraction of Gray and White Matter from the unstructured raw MRI data using automatic feature extraction and deep learning CNN reported with results <85% (in terms of accuracy, sensitivity and specificity) [28,30]. However, using Support Vector Machine (SVM) on unstructured data reported a better specificity in the range 77%–92% table 3. Structural Volume Ratio and Godesic

Length in the study [34] is reported with the highest results <90% (for accuracy, sensitivity and specificity) which is a promising result. One study used neuropsychological data alone on portugal study participants [63] has achieved an accuracy, sensitivity and specificity of 76%, 56% and 76% respectively.

- Parameter performance within populations

However, when mixing ADNI and MILAN study participants using MRI data alone has not shown a better performance with accuracy, sensitivity and specificity of 74%, 75% and 75% respectively which pinpoints towards more research on finding better parameters and learning models while developing algorithms for generalized populations (mixing populations in various geographical locations) as given in Table 3. However, in this experiment researchers used MRI data alone for achieving this. None of the studies in the literature has used multi modalities of data for early prediction of MCI to AD converters for inter continental study participant populations [65]. In addition, the two studies [63,64] conducted on German and Portugal participants has used different modalities for prediction tasks (MRI, CSF for German and Neuro-psychological assessment data for Portuogal). However, the study [64] on German participants with MRI data and CSF performed better than the Neuropsychological data with accuracy, sensitivity and specificity of 82%, 85% and 90% respectively.

- Performance with respect to follow up period:

Following 8 follow-up data for 6 months has achieved a high accuracy of 92% which is a slight increase in 2% accuracy as compared to following 6 month follow up data for 6 months using MRI data (3 years) as given in table 3. In general, the better results are reported using small duration of follow up periods (6 months in the study) as compared to 1 year as given in table 3.

- Performance with respect to the range of MCI converters predicted

The accuracy, sensitivity and specificity for 6 month conversions are 74%, 81% and 71% respectively [49] as given in table 3. In general, better results are reported for 3 year MCI to AD converters with 4 studies reported with accuracy, sensitivity and specificity 85% [36,40,42,47] as given in table 3 respectively.

A full description of the characteristics of data is given in table 3.

*3.4.5.1. Research challenges.* Following are some the major research challenges that need to be addressed by the researchers:

- Identification of the correct and exact bio-markers from the neuroimaging data is a challenging task for the researchers. This task becomes even more difficult when it comes to a specific population. For instance, the researchers face a challenging task while identifying the most precise bio-markers that are responsible for classifying the MCI to AD converters and non MCI to AD converters.
- The researchers also face a challenging task for finding out the relevant features that are responsible for classifying the MCI to AD converters and non-MCI to AD converters from a set of multimodal features. Finding the important and relevant features from the multiple-modal features is also a challenging task for the researchers.

The finding of quick MCI to AD converters, say within 6 months to 1 year is also a challenging task. This is a challenging task because there is an urgency required among the research community for immediately finding the quick MCI to AD converters and it is required to develop very good longitudinal models for addressing this.

3.4.5.2. Future directions. Following are the future directions where the

#### Table 3

SVM-support vector machine, ICV-Intracranial Volume, CNN-convolutional neural networks, ROI-Region of interest, SAE-Spatial auto encoder, GM-Gray matter, ROI-Region of interest, ICA-Independent component analysis, MLP-Multilayer perceptron, BLR-Binary logistic regression.

Reference	Country	Single Modality Description	Multiple Modality Description	Number and Duration of Follow up Data	MCI Range	Machine Learning	Deep Learning	Ensemble Learning	Result
[27]	USA,	1	Х	6	Unclear	1	х	х	Accuracy
	Canada	MRI		6 months	SVM				71% Sensitivit <u></u> 96%
		Hippocampal Volume, ICV, Entorhinal Volume No Feature Selection							Specificit 53%
28]	USA, Canada	1	X	3	1.5 years	х	1	1	Accuracy 74%
	Callaua	MRI		6 months			CNN		Sensitivit 70%%
		93 ROI Automatic Feature Selection							Specificit 78%
29]	USA, Canada	1	X	6	3 years	1	1	1	Accuracy 76%
		MRI Whole Patches in		6 months		Regression	CNN		Sensitivit 42% Specificit
		Image Automatic Feature							82%
[30]	USA,	Selection ✓	X	6	1 year	х	1	1	Accuracy
	Canada	MRI Patches of Gray and White Matter		6 months					82% Sensitivit 81%
		Regions Automatic Feature Selection					SAE		Specificit 82%
31]	USA, Canada	1	X	4	3 year	1	х	х	Accuracy 83%
	Gundud	PET Metabolic Intensity		6 months		SVM			Sensitivit 87% Specificit
		Values Automatic Feature Selection							78%
[32]	USA, Canada	1	Х	6	3 years	1	х	1	Accuracy 73%
	Gallaua	MRI		6 months		SVM			Sensitivit 69%
		93 ROI GM	Automatic Feature						Specificit 77%
[33]	USA,	Х	Selection ✓	6 month	3 years	1	x	x	Accuracy
-	Canada		MRI (Temporal Gyrus, Hippocampus), PET (Both		Ĵ	Cox Models			84% Sensitivit 86%
			ICA)	Clinical Variables No Feature Selection					Specificit 82%
[34]	USA,	1	х	6	3 year	1	х	х	Accuracy
	Canada	MRI		6 months		SVM			92% Sensitivit 95%
			Structural Volume Ratio, Godesic Length No Feature Selection						Specificit 90%
[35]	USA, Canada	1	x	6	Unclear	1	Х	х	Accuracy 76%
		MRI (Cortical thickness	Hippocampus volume)		6 months Not	SVM			Sensitivit 70% Specificit
			No Feature Selection		Continous				81%
36]	USA, Canada	1	X	6	3 years	$\checkmark$	х	х	Accuracy 88%

# Table 3 (continued)

	Country	Single Modality Description	Multiple Modality Description	Number and Duration of Follow up Data	MCI Range	Machine Learning	Deep Learning	Ensemble Learning	Result
		MRI		6 months					Sensitivit 86%
			Amygdala distance No Feature Selection						Specificit 90%
	USA, Canada	х	1	6	1 year	1	X	1	Accuracy 90%
			Structural MRI			SVM			Sensitivit 86%
			FDG-PET (Patch Volume, Mean Intensity of GM ROI's) Automatic Feature Selection	6 months					Specificit 83%
[38]	USA, Canada	1	х	8	1 year	1	х	Х	Accuracy 92%
	Callaua	MRI		6 months		SVM			92% Sensitivit 93%
		Gray matter regions Automatic Feature Selection							Specificit 92%
[39]	USA, Canada	1	х	6	Unclear	1	х	1	Accuracy 79%
	North America	MRI		6 months		Logistic Regression			Sensitivit 87%
		Selection of Voxels Automatic Feature Selection				C			Specificit 73%
[41]	USA, Canada North	x	✓ MRI, PET, CSF	3 6 months	1.5 years	✓ SVM, SVR	X	1	Accuracy 74% Sensitivit
	America		MRI - GM 93 ROI PET - Mean Intensity 93						54% Specificit 88%
			ROI CSF (TAU, A-BETA) Automatic Feature Selection for MRI, PET						
[40]	USA, Canada	X	✓	6	3 years	1	х	1	Accuracy 89%
	North America		MRI, Neuropsychological	6 months		Sparse Learning			Sensitivit 89%
			MRI-Hippocampus, Ventricles MTL, Entorhinal, Fusiform NM- ADASCog, MMSE No Feature Selection			Method			Specificit 92%
[42]	USA,	х	✓	5	3 years	1	х	1	Accuracy
	Canada North America		MRI - Medial Temporal, Etorhinal Cortex Neuropsychological Clinical	6 months		SVM			91% Sensitivit 95% Specificit 87%
[64]	Germany	х	No Feature Selection ✓	3	1 year	1	x	x	Accuracy 82%
	Europe		MRI-Hippocampal Volume	12 months		SVM			82% Sensitivit 85%
			CSF - Tau, A-Beta No Feature Selection						Specificit 70%
[43]	USA, Canada	X	V	6	3 years		X	X	Accuracy 73%
	North America		MRI-Volume based morphometry, Voxel based morphometry PET,CSF	6 months		SVM			Sensitivit 72% Specificit
			Demographics, Medical Medical History, Neuropsychological Automatic Feature						74%
			Selection						

# K.P. Muhammed Niyas and P. Thiyagarajan

# Table 3 (continued)

leference	Country	Single Modality Description	Multiple Modality Description	Number and Duration of Follow up Data	MCI Range	Machine Learning	Deep Learning	Ensemble Learning	Result
	North America		rs-fMRI,MRI rs-fMRI- Connectivity	6 months		SVM			Sensitivity 94% Specificity
			matrix for 93 ROI MRI- Free Surfer Features No Feature Selection						100%
45]	USA, Canada	Х	$\checkmark$	6	3 years	1	х	х	Accuracy 78%
	North America		MRI-Atrophy score, Genetic-PHS, MMSE No Feature Selection	6 months		Cox Proportional Models			Sensitivity 79% Specificity 77%
14]	USA, Canada	1	Х	6	3 years	Х	1	х	Accuracy 79%
	North America	MRI Freesurfer Intensity		6 months			CNN		Sensitivit 84% Specificit
		Values Automatic Feature Selection							74%
7]	USA, Canada	Х	V	8	3 years	✓	Х	X	Accuracy 93%
	North America		MRI, Genetic, Neuropsychological Assesement	6 months		Binary Logistic Regression			Sensitivit 86%
			MRI-Hippocampal Volume, Texture Value of Voxels, MMSE, APOE-4 MRI-Automatic Feature Selection Others-No Feature						Specificit 83%
3]	Portuogal	х	Selection ✓	5	5 years	1	x	x	Accuracy
	Europe		Word Recall Test, Cancellation Task Verbal Paired Associate Learning, Cube Draw Digit Span, Raven Progressive Metrics No Feature Selection		1 year	Naive Bayes			76% Sensitivit 56% Specificit 70%
8]	USA, Canada	1	Х	6	3 years	1	х	х	Accuracy 65%
	North America	MRI							Sensitivit 70%
		Cortical Thickness, Surface Area, Volume Sulcal Depth, Gyrus Height Multifeature network, in out network Multifeature network, in out network (Automatic Feature Selection		6 months					Specificit 58%
65]	USA, Canada, Italy	1	X	6	3 years	х	1	x	Accuracy 74%
	North America, Europe	MRI-Gray Matter, White Matter Intensity		6 months					Sensitivit 75%
101	110 4	Automatic Feature Extraction	,	6	6	,	CNN	v	Specificity 75%
19]	USA, Canada North	X	✓ MRI - Freesurfer	6 6 months	6 months	✓ SVM	x	Х	Accuracy 74% Sensitivit
	America		CSF- A Beta 42, Peptide,						81% Specificit

# K.P. Muhammed Niyas and P. Thiyagarajan

# Table 3 (continued)

Reference	Country	Single Modality Description	Multiple Modality Description	Number and Duration of Follow up Data	MCI Range	Machine Learning	Deep Learning	Ensemble Learning	Result
			Neuropsychological- MMSE No Feature Selection						
[50]	USA, Canada	х	✓	6	3 years	1	X	1	Accuracy 78%
	North America		Socio-Demographic Characteristics - Age, Sex, Education Neuropsychological tests - MMSE, ADAS-Cog, RAVLT No Feature Selection	6 months		SVM, KNN, MLP			Sensitivity 77% Specificity 78%
[62]	USA, Canada	MRI	Not Clear	1	1	1	х		Accuracy 67%
	Gallaŭa	Gray Matter Intensity				SVM			Sensitivity 68% Specificity 66%
[61]	USA, Canada	Х	6 months	Unclear	1	X	1	X	Accuracy 74%
	Australia						CNN		Sensitivity 73%
	Australia								Specificity 67%
[60]	USA, Canada North	✓ MBI patabas	Not Clear	1	х	1	X SVM	Х	Accuracy 80% Precision
	America	MRI patches					SV MI		85% Sensitivit 82% Specificit 80%
59]	USA, Canada,	✓	1	2 years	Unclear	1	х	х	Accuracy 99%
	North America	Whole MRI brain image			2	CNN			Sensitivit 98%
					2	attention mechanism			Specificit 100%
[38]	USA, Canada,	1	X	Unclear	Unclear	1	1	x	Accuracy 92%
	North America	MRI patches				SVM	SAE		Sensitivit 93% Specificit
52]	USA, Canada,	1	x	Unclear	Unclear	1	x	x	92% Accuracy 89%
	North America	MRI Hippocampal				SVM			Sensitivit 90%
	- Interret	MRI Ventricles							Specificit 88%
[53]	USA, Canada,	Х	1	Unclear	Unclear	1	1	х	Accuracy 77%
	North America	MRI 93 ROI				BLR	CNN		Sensitivit 79% Specificit
[54]	England,	1	х	Unclear	Unclear	X	1	x	76% Accuracy
	Europe	Whole patches of MRI					CNN		76% Sensitivit 79%
									Specificit 76%
55]	Italy, Europe	1	X	Unclear	Unclear	X	1	x	Accuracy 91%
		Time Frequency EEG data					CNN		Sensitivit 91% Specificit
56]	USA,	EEG data	x	Unclear	Unclear	X	1	X	92% Accuracy
	Canada, North	Whole patches MRI					CNN		88% Sensitivit
	America	Cognitive data					Attention		88% Specificit
							mechanism	<i>.</i> .	88% d on next pag

#### Table 3 (continued)

Reference	Country	Single Modality Description	Multiple Modality Description	Number and Duration of Follow up Data	MCI Range	Machine Learning	Deep Learning	Ensemble Learning	Result
[66]	USA, Canada,	1	х	Unclear	Unclear	1	1	Х	Accuracy 74%
	North America	Whole patches MRI			SVM	CNN		Sensitivity 73%	
		Cognitive data							Specificity 72%
[58]	USA, Canada,	х	1	Unclear	Unclear	1	X	х	Accuracy 72%
	North America	Multimodal brain							Sensitivity 72%
		Cognitive data							Specificity 72%
[65]	USA, Canada,	1	X	Unclear	Unclear	х	1	1	Accuracy 81%
	North America	MRI Patches					CNN + RNN		Sensitivity 82% Specificity
[44]	USA, Canada,	1	х	Unclear	Unclear	x	1	1	81% Accuracy 73%
		MRI brain regions		CNN + RNN			CNN + RNN		Sensitivity 71%
		Hippocampus, Ventricles, Amygdala				Specificity 75%			
[49]	USA, Canada,	✓	х	Unclear	Unclear	X	1	1	Accuracy 73%
	North America	MRI brain regions					CNN + RNN		Sensitivity 82%
		Graph connectivity							Specificity 84%
		93 ROI							Specificity 82%

researchers can focus their attention.

- It will be interesting to see more experiments on a mix of population from various countries and regions around the world. Consequently, this requires a much larger collaboration among the hospitals and doctors around the globe.<sup>2</sup> There is a large scope for research addressing the creation of a model for a generalized population. As there are few studies that addressed the populations apart from USA and Canada, more number of experiments should focus on AD patients in other parts of the world.
- It will be also interesting to analyze the parameters responsible for AD. How does it vary from region to region and from country to country. A research at this level whereby finding the significant parameters responsible for MCI to AD conversions can give more insights.
- Important challenge in the early prediction of MCI to AD conversion is the identification of the parameters that can report good results in the machine learning process. This would pinpoint towards the necessity for a good feature selection technique. This can be made more specific by creating an algorithm that can find the best set of features and by assigning weight to the features.
- Prediction of MCI to AD conversions within a specific small amount of time is also a challenging task. For example, finding the patients who will convert to AD within the next 3 months or 6 months. Consequently, better studies should be conducted to achieve this.

#### 4. Limitation of the study

The limitations of the existing study is as follows:

- The study is conducted on a selected set of literature from 2017 onwards. Our subsequent work will be to focus on comprehensive literature on Alzheimer's detection from 2012 onwards. This is important for the research community because it will help a new researcher to know the evolution of models and feature extraction from the past.
- The study is focused on finding out the ML papers that identify pMCI, sMCI patients which is the most challenging task. However, we would like to extend our study to identifying AD, MCI, and HC patients.
- A detailed explanation of feature extraction methods is not given in this paper. We are planning an in-depth study on the various feature extraction techniques for classifying pMCI and sMCI.
- We are also planning to focus on a detailed explanation of the various deep learning architectures used for detecting pMCI and sMCI patients. Many studies for pMCI and sMCI patients are performed on medical images using deep neural networks for feature extraction.

#### 5. Conclusion

The parameters responsible for distinguishing MCI to AD converters is not specific and is unknown. This emphasizes the need for good feature selection algorithms that can trace out the best set of parameters out of a big dataset. Moreover, this review also found out a drawback of lack of experiments on various hospital setting data of various countries. A large scale co-operation of AD data will help researchers in experimenting with variables of diverse population and designing a model appropriately, say a single model for generalized population within a country, continent. Etc. In short, the current research requires more generalized models. It is also observed that multiple modalities of data is reporting better results also. This focus on selection of distinguishing features from multi-modalities of data and designing a model

<sup>&</sup>lt;sup>2</sup> http://www.gaain.org/.

accordingly. Moreover, finding quick MCI to AD converters (say within 6 months) using less number of follow up data is also a challenging task.

#### Declaration of competing interest

The authors declare that they have no known competing financial interests or personal relationships that could have appeared to influence the work reported in this paper.

#### References

- Lucy Burke, The poetry of dementia: art, ethics and alzheimer's disease in tony harrison's black daisies for the bride, J. Lit. Cult. Disabil. Stud. 1 (1) (2007) 61–73.
- [2] Georgia Lowndes, Greg Savage, Early detection of memory impairment in alzheimer's disease: a neurocognitive perspective on assessment, Neuropsychol. Rev. 17 (3) (2007) 193–202.
- [3] Keliang Pang, Richeng Jiang, Wei Zhang, Zhengyi Yang, Lin-Lin Li, Makoto Shimozawa, Simone Tambaro, Johanna Mayer, Baogui Zhang, Man Li, et al., An app knock-in rat model for alzheimer's disease exhibiting a β and tau pathologies, neuronal death and cognitive impairments, Cell Res. 32 (2) (2022) 157–175.
- [4] Peidong Chen, Zhiwei Shen, Qianqian Wang, Bingna Zhang, Zerui Zhuang, Jiefen Lin, Yuanyu Shen, Yanzhi Chen, Zhuozhi Dai, Renhua Wu, Reduced cerebral glucose uptake in an alzheimer's rat model with glucose-weighted chemical exchange saturation transfer imaging, Front. Aging Neurosci. 13 (2021), 618690.
- [5] Anthoni M. Goodman, Bethany M. Langner, Nateka Jackson, Alex Capri, Lori L. McMahon, Heightened hippocampal β-adrenergic receptor function drives synaptic potentiation and supports learning and memory in the tgf344-ad rat model during prodromal alzheimer's disease, J. Neurosci. 41 (26) (2021) 5747–5761.
- [6] Zhe-yu Zhang, Chun-hu Zhang, Jing-jing Yang, Pan-pan Xu, Peng-ji Yi, Mu-li Hu, Wei-jun Peng, Genome-wide analysis of hippocampal transfer rna-derived small rnas identifies new potential therapeutic targets of bushen tiansui formula against alzheimer's disease, J. Integrat. Med. 19 (2) (2021) 135–143.
- [7] Nicole M. Ralbovsky, Greg S. Fitzgerald, Ewan C. McNay, Igor K. Lednev, Towards development of a novel screening method for identifying alzheimer's disease risk: Raman spectroscopy of blood serum and machine learning, Spectrochim. Acta Mol. Biomol. Spectrosc. 254 (2021), 119603.
- [8] Xiao Chen, Min Zhang, Mukhtar Ahmed, Krishna Mohan Surapaneni, Vishnu Priya Veeraraghavan, Palanisamy Arulselvan, Neuroprotective effects of ononin against the aluminium chloride-induced alzheimer's disease in rats, Saudi J. Biol. Sci. 28 (8) (2021) 4232–4239.
- [9] Bennett P. Leifer, Early diagnosis of alzheimer's disease: clinical and economic benefits, J. Am. Geriatr. Soc. 51 (5s2) (2003) S281–S288.
- [10] Thomas G. Beach, Sarah E. Monsell, Leslie E. Phillips, Kukull Walter, Accuracy of the clinical diagnosis of alzheimer disease at national institute on aging alzheimer disease centers, 2005–2010, J. Neuropathol. Exp. Neurol. 71 (4) (2012) 266–273.
- [11] Hui-Xin Wang, Laura Fratiglioni, Giovanni B. Frisoni, Matti Viitanen, Bengt Winblad, Smoking and the occurrence of alzheimer's disease: cross-sectional and longitudinal data in a population-based study, Am. J. Epidemiol. 149 (7) (1999) 640–644.
- [12] Patrick Doupe, James Faghmous, Sanjay Basu, Machine learning for health services researchers, Value Health 22 (7) (2019) 808–815.
- [13] Tausifa J. Saleem, Mohammad Ahsan Chishti, Exploring the applications of machine learning in healthcare, Int. J. Sensor. Wireless Commun. Control 10 (4) (2020) 458–472.
- [14] Tom C. Russ, G. David Batty, Gena F. Hearnshaw, Candida Fenton, John M. Starr, Geographical variation in dementia: systematic review with meta-analysis, Int. J. Epidemiol. 41 (4) (2012) 1012–1032.
- [15] Liara Rizzi, Idiane Rosset, Matheus Roriz-Cruz, 2014, Global Epidemiology of
- Dementia: Alzheimer's and Vascular Types, BioMed research international, 2014.
  [16] Ronald C. Petersen, Glenn E. Smith, Stephen C. Waring, Robert J. Ivnik, Eric G. Tangalos, Emre Kokmen, Mild cognitive impairment: clinical characterization and outcome. Arch. Neurol. 56 (3) (1999) 303–308.
- [17] Matthew J. Page, David Moher, Joanne E. McKenzie, Introduction to prisma 2020 and implications for research synthesis methodologists, Res. Synth. Methods 13 (2) (2022) 156–163.
- [18] CR Jack Jr et al. Alzheimer's Disease Neuroimaging Initiative Dataset.
- [19] Susanne G. Mueller, Michael W. Weiner, Leon J. Thal, Ronald C. Petersen, Clifford Jack, William Jagust, John Q. Trojanowski, Arthur W. Toga, Laurel Beckett, The alzheimer's disease neuroimaging initiative, Neuroimaging Clinics 15 (4) (2005) 869–877.
- [20] Paul S. Aisen, Ronald C. Petersen, Michael Donohue, Michael W. Weiner, et al., Alzheimer's Disease Neuroimaging Initiative, Alzheimer's disease neuroimaging initiative 2 clinical core: progress and plans, Alzheimer's Dementia 11 (7) (2015) 734–739.
- [21] James D. Doecke, Simon M. Laws, Noel G. Faux, William Wilson, Samantha C. Burnham, Chiou-Peng Lam, Alinda Mondal, Justin Bedo, Ashley I. Bush, Belinda Brown, et al., Blood-based protein biomarkers for diagnosis of alzheimer disease, Arch. Neurol. 69 (10) (2012) 1318–1325.
- [22] Christopher Fowler, Stephanie R. Rainey-Smith, Sabine Bird, Bomke Julia, Pierrick Bourgeat, Belinda M. Brown, Samantha C. Burnham, Ashley I. Bush, Carolyn Chadunow, Steven Collins, et al., Fifteen years of the australian imaging,

biomarkers and lifestyle (aibl) study: progress and observations from 2,359 older adults spanning the spectrum from cognitive normality to alzheimer's disease, J. Alzheimer's Dis. Rep. 5 (1) (2021) 443–468.

- [23] Pamela J La Montagne, Tammie LS. Benzinger, John C. Morris, Sarah Keefe, Russ Hornbeck, Chengjie Xiong, Elizabeth Grant, Jason Hassenstab, Krista Moulder, Andrei G. Vlassenko, et al., Oasis-3: Longitudinal Neuroimaging, Clinical, and Cognitive Dataset for Normal Aging and Alzheimer Disease, 2019. MedRxiv.
- [24] Xin Li, Shaomin Liu, Qin Xiao, Mingguo Ma, Rui Jin, Tao Che, Weizhen Wang, Xiaoli Hu, Ziwei Xu, Jianguang Wen, et al., A multiscale dataset for understanding complex eco-hydrological processes in a heterogeneous oasis system, Sci. Data 4 (1) (2017) 1–11.
- [25] Yunus Miah, Chowdhury Nazia Enam Prima, Sharmeen Jahan Seema, Mufti Mahmud, M Shamim Kaiser, Performance comparison of machine learning techniques in identifying dementia from open access clinical datasets, in: Advances on Smart and Soft Computing, 2021, pp. 79–89. Springer.
- [26] Aaron R. Ritter, Gabriel C. Leger, Justin B. Miller, Sarah J. Banks, Neuropsychological testing in pathologically verified alzheimer's disease and frontotemporal dementia: how well do the uniform data set measures differentiate between diseases? Alzheimer Dis. Assoc. Disord. 31 (3) (2017) 187.
- [27] Marta Gomez-Sancho, Jussi Tohka, Vanessa Gomez-Verdejo, et al., Alzheimer's Disease Neuroimaging Initiative, Comparison of feature representations in mribased mci-to-ad conversion prediction, Magn. Reson. Imaging 50 (2018) 84–95.
- [28] Heung-Il Suk, Seong-Whan Lee, Dinggang Shen, et al., Alzheimer's Disease Neuroimaging Initiative, Deep ensemble learning of sparse regression models for brain disease diagnosis, Med. Image Anal. 37 (2017) 101–113.
- [29] Mingxia Liu, Jun Zhang, Ehsan Adeli, Dinggang Shen, Landmark-based deep multiinstance learning for brain disease diagnosis, Med. Image Anal. 43 (2018) 157–168.
- [30] Donghuan Lu, Karteek Popuri, Gavin Weiguang Ding, Rakesh Balachandar, Mirza Faisal Beg, et al., Alzheimer's Disease Neuroimaging Initiative, Multiscale deep neural network based analysis of fdg-pet images for the early diagnosis of alzheimer's disease, Med. Image Anal. 46 (2018) 26–34.
- [31] Yu Zhao, Zhijun Yao, Weihao Zheng, Jing Yang, Zhijie Ding, Mi Li, Shengfu Lu, Predicting mci progression with individual metabolic network based on longitudinal fdg-pet, 1894–1899, in: 2017 IEEE International Conference on Bioinformatics and Biomedicine (BIBM)IEEE, 2017.
- [32] Bo Cheng, Mingxia Liu, Dinggang Shen, Zuoyong Li, Daoqiang Zhang, Alzheimer's Disease Neuroimaging Initiative, et al., Multi-domain transfer learning for early diagnosis of alzheimer's disease, Neuroinformatics 15 (2) (2017) 115–132.
- [33] Ke Liu, Kewei Chen, Li Yao, Xiaojuan Guo, Prediction of mild cognitive impairment conversion using a combination of independent component analysis and the cox model, Front. Hum. Neurosci. 11 (2017) 33.
- [34] Zhuo Sun, Martijn van de Giessen, Boudewijn PF. Lelieveldt, Marius Staring, Detection of conversion from mild cognitive impairment to alzheimer's disease using longitudinal brain mri, Front. Neuroinf. 11 (16) (2017).
- [35] Giovana Gavidia-Bovadilla, Samir Kanaan-Izquierdo, María Mataró-Serrat, Alexandre Perera-Lluna, et al., Alzheimer's Disease Neuroimaging Initiative, Early prediction of alzheimer's disease using null longitudinal model-based classifiers, PLoS One 12 (1) (2017), e0168011.
- [36] Xiaojing Long, Lifang Chen, Chunxiang Jiang, Lijuan Zhang, Alzheimer's Disease Neuroimaging Initiative, et al., Prediction and classification of alzheimer disease based on quantification of mri deformation, PLoS One 12 (3) (2017), e0173372.
- [37] Donghuan Lu, Karteek Popuri, Gavin Weiguang Ding, Rakesh Balachandar, Mirza Faisal Beg, Multimodal and multiscale deep neural networks for the early diagnosis of alzheimer's disease using structural mr and fdg-pet images, Sci. Rep. 8 (1) (2018) 5697.
- [38] Ting Shen, Yupeng Li, Ping Wu, Chuantao Zuo, Zhuangzhi Yan, Decision supporting model for one-year conversion probability from mci to ad using cnn and svm, in: 2018 40th Annual International Conference of the IEEE Engineering in Medicine and Biology Society (EMBC), IEEE, 2018, pp. 738–741.
- [39] Meiyan Huang, Wei Yang, Qianjin Feng, Wufan Chen, Michael W. Weiner, Aisen Paul, Ronald Petersen, Clifford R. Jack Jr., William Jagust, John Q. Trojanowki, et al., Longitudinal measurement and hierarchical classification framework for the prediction of alzheimer's disease, Sci. Rep. 7 (2017), 39880.
- [40] Sidra Minhas, Aasia Khanum, Farhan Riaz, Atif Alvi, Shoab Ahmed Khan, A nonparametric approach for mild cognitive impairment to ad conversion prediction: results on longitudinal data, IEEE J. Biomed. Health Informat. 21 (5) (2016) 1403–1410.
- [41] Baiying Lei, Peng Yang, Tianfu Wang, Siping Chen, Ni Dong, Relational-regularized discriminative sparse learning for alzheimer's disease diagnosis, IEEE Trans. Cybern. 47 (4) (2017) 1102–1113.
- [42] Massimiliano Grassi, Giampaolo Perna, Daniela Caldirola, Koen Schruers, Ranjan Duara, David A. Loewenstein, A clinically-translatable machine learning algorithm for the prediction of alzheimer's disease conversion in individuals with mild and premild cognitive impairment, J. Alzheim. Dis. 61 (4) (2018) 1555–1573.
- [43] Kerstin Ritter, Julia Schumacher, Martin Weygandt, Ralph Buchert, Carsten Allefeld, John-Dylan Haynes, et al., Alzheimer's Disease Neuroimaging Initiative, Multimodal prediction of conversion to alzheimer's disease based on incomplete biomarkers, Alzheimer's Dementia: Diagn. Assess. Dis. Monitor. 1 (2) (2015) 206–215.
- [44] Weiming Lin, Tong Tong, Qinquan Gao, Di Guo, Xiaofeng Du, Yonggui Yang, Gang Guo, Min Xiao, Min Du, Xiaobo Qu, et al., Convolutional neural networksbased mri image analysis for the alzheimer's disease prediction from mild cognitive impairment, Front. Neurosci. 12 (2018).

#### K.P. Muhammed Niyas and P. Thiyagarajan

- [45] Karolina Kauppi, Chun Chieh Fan, Linda K. McEvoy, Dominic Holland, Chin Hong Tan, Chi-Hua Chen, Ole A. Andreassen, Rahul S. Desikan, M Dale Anders, et al., Alzheimer's Disease Neuroimaging Initiative, Combining polygenic hazard score with volumetric mri and cognitive measures improves prediction of progression from mild cognitive impairment to alzheimer's disease, Front. Neurosci. 12 (2018).
- [46] Seyed Hani Hojjati, Ata Ebrahimzadeh, Khazaee Ali, Abbas Babajani-Feremi, et al., Alzheimer's Disease Neuroimaging Initiative, Predicting conversion from mci to ad by integrating rs-fmri and structural mri, Comput. Biol. Med. 102 (2018) 30–39.
- [47] C Luk Collin, Abdullah Ishaque, Muhammad Khan, Ta Daniel, Sneha Chenji, Yee-Hong Yang, Eurich Dean, Sanjay Kalra, et al., Alzheimer's Disease Neuroimaging Initiative, Alzheimer's disease: 3-dimensional mri texture for prediction of conversion from mild cognitive impairment, Alzheimer's Dementia: Diagn. Assess. Dis. Monitor. 10 (2018) 755–763.
- [48] Weihao Zheng, Zhijun Yao, Yuanwei Xie, Jin Fan, Bin Hu, Identification of alzheimer's disease and mild cognitive impairment using networks constructed based on multiple morphological brain features, Biol. Psychiatr.: Cognit. Neurosci. Neuroimag. 3 (10) (2018) 887–897.
- [49] Garam Lee, Kwangsik Nho, Byungkon Kang, Kyung-Ah Sohn, Dokyoon Kim, Predicting alzheimer's disease progression using multi-modal deep learning approach, Sci. Rep. 9 (1) (2019) 1952.
- [50] Massimiliano Grassi, Nadine Rouleaux, Daniela Caldirola, David Loewenstein, Koen Schruers, Giampaolo Perna, Michel Dumontier, Alzheimer's Disease Neuroimaging Initiative, et al., A novel ensemble-based machine learning algorithm to predict the conversion from mild cognitive impairment to alzheimer's disease using socio-demographic characteristics, clinical information, and neuropsychological measures, Front. Neurol. 10 (2019).
- [51] Rémi Cuingnet, Emilie Gerardin, Jérôme Tessieras, Guillaume Auzias, Stéphane Lehéricy, Marie-Odile Habert, Marie Chupin, Habib Benali, Olivier Colliot, Alzheimer's Disease Neuroimaging Initiative, et al., Automatic classification of patients with alzheimer's disease from structural mri: a comparison of ten methods using the adni database, Neuroimage 56 (2) (2011) 766–781.
- [52] Paolo Maria Rossini, Francesca Miraglia, Fabrizio Vecchio, Early Dementia Diagnosis, Mci-To-Dementia Risk Prediction, and the Role of Machine Learning Methods for Feature Extraction from Integrated Biomarkers, in Particular for Eeg Signal Analysis, Alzheimer's & Dementia, 2022.
- [53] Patricio Andres Donnelly-Kehoe, Guido Orlando Pascariello, Juan Carlos Gómez, et al., Alzheimers Disease Neuroimaging Initiative, Looking for alzheimer's disease morphometric signatures using machine learning techniques, J. Neurosci. Methods 302 (2018) 24–34.
- [54] Fei Gao, Hyunsoo Yoon, Yanzhe Xu, Dhruman Goradia, Ji Luo, Teresa Wu, Yi Su, et al., Alzheimer's Disease Neuroimaging Initiative, Ad-net: age-adjust neural network for improved mci to ad conversion prediction, Neuroimage: Clinical 27 (2020), 102290.
- [55] Cosimo Ieracitano, Nadia Mammone, Amir Hussain, Francesco C. Morabito, A novel multi-modal machine learning based approach for automatic classification of eeg recordings in dementia, Neural Network. 123 (2020) 176–190.

#### International Journal of Intelligent Networks 4 (2023) 74-88

- [56] Ramon Casanova, Ryan T. Barnard, Sarah A. Gaussoin, Saldana Santiago, Kathleen M. Hayden, JoAnn E. Manson, Robert B. Wallace, Stephen R. Rapp, Susan M. Resnick, Mark A. Espeland, et al., Using high-dimensional machine learning methods to estimate an anatomical risk factor for alzheimer's disease across imaging databases, Neuroimage 183 (2018) 401–411.
- [57] Deniz Sezin Ayvaz, Inci M. Baytas, Investigating conversion from mild cognitive impairment to alzheimer's disease using latent space manipulation, in: arXiv Preprint arXiv:2111.08794, 2021.
- [58] Yiran Wei, Stephen J. Price, Carola-Bibiane Schönlieb, Chao Li, Predicting conversion of mild cognitive impairment to alzheimer's disease, in: arXiv Preprint arXiv:2203.04725, 2022.
- [59] Marianna Inglese, Neva Patel, Kristofer Linton-Reid, Flavia Loreto, Zarni Win, Richard J. Perry, Christopher Carswell, Matthew Grech-Sollars, William R. Crum, Haonan Lu, et al., A predictive model using the mesoscopic architecture of the living brain to detect alzheimer's disease, Commun. Med. 2 (1) (2022) 1–16.
- [60] Juan E. Arco, Javier Ramírez, Juan M. Górriz, María Ruz, et al., Alzheimer's Disease Neuroimaging Initiative, Data fusion based on searchlight analysis for the prediction of alzheimer's disease, Expert Syst. Appl. 185 (2021), 115549.
- [61] Samaneh Abolpour Mofrad, Arvid Lundervold, Alexander Selvikvåg Lundervold, et al., Alzheimer's Disease Neuroimaging Initiative, A predictive framework based on brain volume trajectories enabling early detection of alzheimer's disease, Comput. Med. Imag. Graph. 90 (2021), 101910.
- [62] Esther E. Bron, Štefan Klein, Janne M. Papma, Lize C. Jiskoot, Vikram Venkatraghavan, Linders Jara, Pauline Aalten, Peter Paul De Deyn, Geert Jan Biessels, Jurgen AHR. Claassen, et al., Cross-cohort generalizability of deep and conventional machine learning for mri-based diagnosis and prediction of alzheimer's disease, Neuroimage: Clinical 31 (2021), 102712.
- [63] Telma Pereira, Luís Lemos, Sandra Cardoso, Dina Silva, Ana Rodrigues, Isabel Santana, Alexandre de Mendonça, Manuela Guerreiro, Sara C. Madeira, Predicting progression of mild cognitive impairment to dementia using neuropsychological data: a supervised learning approach using time windows, BMC Med. Inf. Decis. Making 17 (1) (2017) 110.
- [64] Lutz Frölich, Oliver Peters, Piotr Lewczuk, Oliver Gruber, Stefan J. Teipel, Hermann J. Gertz, Holger Jahn, Jessen Frank, Kurz Alexander, Christian Luckhaus, et al., Incremental value of biomarker combinations to predict progression of mild cognitive impairment to alzheimer's dementia, Alzheimer's Res. Ther. 9 (1) (2017) 84.
- [65] Silvia Basaia, Federica Agosta, Luca Wagner, Elisa Canu, Giuseppe Magnani, Roberto Santangelo, Massimo Filippi, et al., Alzheimer's Disease Neuroimaging Initiative, Automated classification of alzheimer's disease and mild cognitive impairment using a single mri and deep neural networks, Neuroimage: Clinical 21 (2019), 101645.
- [66] Fujia Ren, Chenhui Yang, Y.A. Nanehkaran, Mri-based model for mci conversion using deep zero-shot transfer learning, J. Supercomput. (2022) 1–19.
- [67] Manhua Liu, Danni Cheng, Weiwu Yan, Alzheimer's Disease Neuroimaging Initiative, Classification of alzheimer's disease by combination of convolutional and recurrent neural networks using fdg-pet images, Front. Neuroinf. 12 (2018) 35.